Author's response to reviews

Title: Information needs of health workers in developing countries - a literature review with a focus on Africa

Authors:

Neil M Pakenham-Walsh (neil.pakenham-walsh@ghi-net.org)
Frederick O Bukachi (fred.bukachi@ghi-net.org)

Version: 2 Date: 5 March 2009

Author's response to reviews:

Attached is the revised copy of our manuscript (two files: text plus Table 1), addressing the issues raised by the reviewers of the first submission:

1. The original text indicated two purposes. The second purpose has been removed, and the first has been revised, as recommended by one of the reviewers.

2. The abstract has been structured as recommended by the Journal.

3. Formatting now conforms with BioMed Central's instructions.

With thanks,

Neil Pakenham-Walsh & Frederick Bukachi